Author's response to reviews

Title: An analysis of risk factors of non-fatal drowning among children in rural areas of Guangdong Province, China: a case-control study

Authors:

Wen Jun Ma (mwj68@tom.com)
Shao Ping Nie (Nieszhp@163.com)
Hao Feng Xu (xhf62@163.com)
Yan Jun Xu (gdxj05@21cn.com)
Xiu Ling Song (songxl918@163.com)
Qiao Zhi Guo (gqz66@yahoo.cn)
Yu Run Zhang (gdncc@126.com)

Version: 7 Date: 20 March 2010

Author's response to reviews:

Dear editor,

I only made a very minor revision after getting through the manuscript.